Author’s response to reviews

Title: Inadequate vancomycin therapy in term and preterm neonates: a retrospective analysis of trough serum concentrations in relation to minimal inhibitory concentrations

Authors:

Fleur Sinkeler (f.sinkeler@amc.uva.nl)
Timo de Haan (t.r.dehaan@amc.uva.nl)
Caspar Hodiamont (c.j.hodiamont@amc.uva.nl)
Dasja Pajkrt (d.pajkrt@amc.uva.nl)
Yuma Bijleveldt (y.a.bijleveld@amc.uva.nl)
Ron Mathot (r.mathot@amc.uva.nl)

Version: 2
Date: 25 February 2014

Author’s response to reviews:

Dear Mrs Olino,

In reaction to your e-mail:

1. Author names and email addresses have been corrected.
2. A ‘Competing interests’ section has been added to the manuscript.

Kind regards,
Ron Mathôt